Author’s response to reviews

Title: Characterization of a calcified intra-cardiac pseudocyst of the mitral valve by magnetic resonance imaging including T1 and T2 mapping

Authors:

Ursula Reiter (ursula.reiter@klinikum-graz.at)
Gert Reiter (gert.reiter@siemens.com)
Martin Asslaber (martin.asslaber@medunigraz.at)
Drago Dacar (drago.dacar@medunigraz.at)
Ralph Maderthaner (ralph.maderthaner@klinikum-graz.at)
Josepha Binder (josepha.binder@klinikum-graz.at)
Andreas Greiser (andreas.greiser@siemens.com)
Meinrad Beer (Meinrad.Beer@uniklinik-ulm.de)
Michael Fuchsjaeger (michael.fuchsjaeger@medunigraz.at)

Version: 2
Date: 19 November 2013

Author’s response to reviews: see over
Dear Editorial Board of BMC Cardiovascular Disorders,

It is with great pleasure that we resubmit our manuscript entitled

Characterization of a calcified intra-cardiac pseudocyst of the mitral valve
by magnetic resonance imaging including T1 and T2 mapping

for review and possible publication as Case Report in BMC Cardiovascular Disorders.

Unfortunately we entered an unstructured abstract in the submission system. We corrected the abstract in the submission system, which is now consistent with the abstract in the manuscript.

Best regards,
Ursula Reiter

Point-by-point description of changes:

1) Please note that the abstract in your manuscript file differs from the abstract entered in the submission system - please correct so they are consistent with each other.

We corrected the abstract in the submission system

(2) Please provide an abstract formatted according to the guidelines for authors <http://www.biomedcentral.com/info/ifora/abstracts>. Potential referees will be asked to review the manuscript having seen only the title and abstract, so it is important that these are both informative and concise.

We structured the abstract according to the guidelines for case reports.